Author’s response to reviews

Title: West End Walkers 65+ A randomised controlled trial of a primary care-based walking intervention for older adults: Study rationale and design

Authors:

Freya MacMillan (freya.macmillan@strath.ac.uk)
Claire Fitzsimons (claire.fitzsimons@strath.ac.uk)
Karen Black (karen.black@strath.ac.uk)
Malcolm H Granat (malcolm.granat@gcu.ac.uk)
P Margaret Grant (M.Grant@gcu.ac.uk)
Madeleine Grealy (m.grealy@strath.ac.uk)
Hazel Macdonald (hazel.macdonald@nhs.net)
Alex McConnachie (Alex.McConnachie@glasgow.ac.uk)
David A Rowe (david.rowe@strath.ac.uk)
Rebecca Shaw (Rebecca.Shaw@glasgow.ac.uk)
Dawn A Skelton (Dawn.Skelton@gcu.ac.uk)
Nanette Mutrie (nanette.mutrie@strath.ac.uk)

Version: 6 Date: 15 February 2011

Author’s response to reviews:

Please find attached a final revision of our submission of the manuscript entitled: ‘West End Walkers 65+
A randomised controlled trial of a primary care-based walking intervention for older adults: Study rationale and design.’

The following change has been made to the last sentence of the first paragraph on page 9:-

The word ‘a’ has been inserted before ‘graduated pedometer-based walking programme’

Also in the ‘figure 1’ document the words 40-60 minutes in the top box are not showing in full and I would appreciate if the box could be made slightly bigger.

After these changes have been made we confirm that the manuscript is acceptable for publication.

On behalf of the authors,

Nanette Mutrie